
-----------------------------------------------------------------------------
          Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
       Mon Aug 11 16:27:48 GMT 2014
Year: 14  Doy: 223
Observer: berkey
WEATHER COMMENT: berkey: Mon Aug 11 16:27:52 GMT 2014
temp 40f, wind 6mph from the south, clear skies, boardwalk was wet so I guess it was rainy or foggy last night.
___end___
GENERAL OBSERVATORY COMMENT BY berkey: Mon Aug 11 16:34:27 GMT 2014
Per some remarks in yesterdays log.
mlsoserver inaccessable.  Looks like the network switch crashed.  All lights were illumnated with zero activity.  Restarti
ng cleared this up.
comp keyboard not working:  kvm usb keyboard cable was unplugged.  After plugging it back in comp came back.
Pspt dos box crashed rebooting fixed it.
Pspt camera/camera chiller timer had the wrong time.  Time was reset
___end___
Mon Aug 11 16:53:26 GMT 2014 COMP Start Patrol
Mon Aug 11 17:02:49 GMT 2014: PSPT Start Patrol
Mon Aug 11 17:08:37 GMT 2014: PSPT Abort Patrol
Mon Aug 11 17:08:41 GMT 2014: PSPT Abort Patrol
Mon Aug 11 17:09:05 GMT 2014 KCOR Start Synoptic Patrol
Mon Aug 11 17:13:49 GMT 2014 KCOR End Patrol
Mon Aug 11 17:23:54 GMT 2014: PSPT Start Patrol
**CoMP PROBLEM COMMENT BY berkey **: Mon Aug 11 17:28:23 GMT 2014
Comp data drive died. Swapped with a spare.
___end___
Mon Aug 11 17:29:37 GMT 2014 KCOR Start Synoptic Patrol
Mon Aug 11 17:35:08 GMT 2014 COMP Start Patrol
Mon Aug 11 17:41:26 GMT 2014: PSPT Start Patrol
Mon Aug 11 17:41:47 GMT 2014: PSPT Abort Patrol
Mon Aug 11 17:41:52 GMT 2014Mon Aug 11 17:50:08 GMT 2014: PSPT Start Patrol
Mon Aug 11 18:24:25 GMT 2014 KCOR End Patrol
Mon Aug 11 18:24:25 GMT 2014 KCOR Start Calibration script: c:\kcor\mlso-calibration22deg.ini
Mon Aug 11 18:40:17 GMT 2014 KCOR End Calibration Script
Mon Aug 11 18:40:34 GMT 2014 KCOR Start Synoptic Patrol
Mon Aug 11 18:40:34 GMT 2014 KCOR Start Synoptic Patrol
Mon Aug 11 19:54:45 GMT 2014 KCOR End Patrol
Mon Aug 11 19:55:03 GMT 2014 COMP End Patrol
Mon Aug 11 21:01:32 GMT 2014: PSPT Abort Patrol
Mon Aug 11 21:01:36 GMT 2014: PSPT Abort Patrol
GENERAL OBSERVATORY COMMENT BY berkey: Mon Aug 11 21:06:21 GMT 2014
thick clouds
___end___
Mon Aug 11 21:18:18 GMT 2014: PSPT Start Patrol
Mon Aug 11 22:35:37 GMT 2014: PSPT Abort Patrol
Mon Aug 11 22:35:41 GMT 2014: PSPT Abort Patrol
CoMP COMMENT BY berkey: Mon Aug 11 22:41:11 GMT 2014



CoMP O1 washed.
___end___
GENERAL OBSERVATORY COMMENT BY berkey: Mon Aug 11 23:51:14 GMT 2014
Little bit of fog
___end___
KCOR COMMENT BY berkey: Tue Aug 12 01:07:54 GMT 2014
It looked like kcor wans’t updated the t-cam/r-cam focus postion. A check was added to the mc4u controller to check the t/
r cam postions and write it to the fits header.
___end___
KCOR COMMENT BY berkey: Tue Aug 12 01:13:30 GMT 2014
Kcor lens inspected.  Lots of small bug and dust.  It was blown off hopefully we wont have bulleyes in tomorrows data.
___end___
GENERAL OBSERVATORY COMMENT BY berkey: Tue Aug 12 01:32:33 GMT 2014
wireless access point seems dead. a new one was d should be at the observatory fridau. 
___end___


